Author’s response to reviews

Title: Developing a tablet computer based application ('App') to measure self-reported alcohol consumption in Indigenous Australians

Authors:

KS Lee (kylie.lee@sydney.edu.au)
Scott Wilson (scott@adac.org.au)
Jimmy Perry (jimmy@adac.org.au)
Robin Room (r.room@latrobe.edu.au)
Sarah Callinan (s.callinan@latrobe.edu.au)
Robert Assan (Robert_Assan@health.qld.gov.au)
Noel Hayman (noel.hayman@health.qld.gov.au)
Tanya Chikritzhs (t.n.chikritzhs@curtin.edu.au)
Dennis Gray (d.gray@curtin.edu.au)
Edward Wilkes (e.wilkes@curtin.edu.au)
Peter Jack (peter.jack@sswahs.nsw.gov.au)
Katherine Conigrave (kate.conigrave@sydney.edu.au)

Version: 2 Date: 28 Dec 2017

Author’s response to reviews:

see attached letter